Author's response to reviews

Title: Rare recurrence of a rare ovarian stromal tumor with luteinized cells: a case report 2.

Authors:

- TAZI EL MEHDI (moulay.elmehdi@yahoo.fr)
- ISMAIL ESSADI (ismail@yahoo.fr)
- HIND M' Rabti (hind@yahoo.fr)
- HASSAN ERRIHANI (hassan@yahoo.fr)

Version: 4 Date: 17 May 2011

Author's response to reviews: see over
Dear Sir,

I do all the modifications.

1. I restructured the title and the abstract.
2. I include the ethnicity of the patient.
3. I restructured the conclusion
4. I include the consent of the patient
5. The Competing Interests statement is above Author Contributions
6. The text on the right hand side of the Figure 2 was blocked out

Sincerely yours

Dr El Mehdi Tazi